Page 1 of 65



ENG-STIPEL 10001:2025

Central Scheme

Central scheme for the conformity assessment of persons dealing with safety risks when working in energy engineering

Version 2025 - Definitive

Stichting Persoonscertificatie Energietechniek

All rights reserved. Nothing in this publication may be reproduced, stored in an automated database, or made public, in any form or in any way, either electronically, mechanically, by photocopying, recording, or any other way, without STIPEL's prior written consent.

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



Content

INTRO	DDUCTION	. 8
1	SUBJECT AND SCOPE	10
2	NORMATIVE REFERENCES	10
3	TERMS AND DEFINITIONS	11
PART	1: REQUIREMENTS FOR THE CERTIFICATION PROCESS	15
4	CONTENT OF THE CENTRAL SCHEME	15
4.1	GENERAL	15
4.2	PURPOSE OF THE CENTRAL SCHEMA	15
4.2.1	OPERATIONAL SAFETY	15
4.2.2	HAZARDS	15
4.3	OBJECT OF CERTIFICATION	17
4.3.1	ELEMENTS TO DETERMINE THE SCOPE	17
4.3.2	DUTIES AND RESPONSIBILITIES	18
4.3.3	AUTHORISED PROFILES	19
4.4	CONFORMITY ASSESMENT	19
4.4.1	CONFORMITY ASSESSMENT NEN 3140 AND NEN 3840	19
4.4.1.1	1 PREREQUISITES	19
4.4.1.2	2 CERTIFICATION TEST: THEORETICAL EXAMINATION	20
4.4.1.3	3 CERTIFICATIETEST: PRACTICAL EXAMINATION	20
4.4.1.4	4 RECERTIFICATION	20
4.4.2	CONFORMITY ASSESSMENT BEI BLS, BEI BHS AND VIAG	21
4.4.2.1	1 PREREQUISITES	21
4.4.2.2	2 CERTIFICATION TEST: THEORETICAL EXAMINATION	21
4.4.2.3	3 CERTIFICATION TEST: PRACTICAL EXAMINATION	22
4.4.2.4	4 RECERTIFICATION	22

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



PAR	T 2: REQUIREMENTS FOR THE EXAMINATION PROCESS
5	GENERAL REQUIREMENTS THEORETICAL AND PRACTICAL EXAMS
5.1	GENERAL REQUIREMENTS THEORETICAL EXAMS
5.1.1	CONDUCTING THEORETICAL EXAMINATIONS
5.1.2	TEST FORM
5.2	GENERAL REQUIREMENTS PRACTICAL EXAMS
5.3	OTHER GENERAL REQUIREMENTS EXAMS 24
5.3.1	LANGUAGE OF THE EXAM
5.3.2	USE OF REFERENCE MATERIALS AND DICTIONARIES
5.3.3	TAKING THEORETICAL- AND PRACTICAL EXAMINATION INDIVIDUALLY
5.3.4	REMOTE EXAMS
5.4 3840	ASSESSMENT OF THEORETICAL AND PRACTICAL EXAMS NEN 3140 AND NEN 26
5.4.1 3840	ASSESMENT OF THEORETICAL AND PRACTICAL EXAMS NEN 3140 AND NEN 26
5.4.2	ASSESSMENT OF THEORETICAL AND PRACTICAL EXAMS
5.4.3	ASSESSMENT OF THEORETICAL AND PRACTICAL EXAMS BEI BHS
5.4.4	ASSESSMENT OF THEORETICAL AND PRACTICAL EXAMINATION VIAG
6	GENERAL PERSONNEL REQUIREMENTS
6.1	EXAMINERS
6.2	PERSONS WHO EVALUATE THE EXAM RESULTS
7 3140	REQUIREMENTS FOR PRACTICAL LOCATIONS UNDER THE SCOPE OF NEN 29
7.1	GENERAL INFORMATION
7.2	SAFETY AND TRUE TO LIFE SITUATIONS
7.3	ELEMENTARY SET-UP REQUIREMENTS
7.4	AVAILABLE MATERIALS

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



8 OF N	REQUIREMENTS FOR PRACTICAL LOCATIONS FOR THE ARE OF APPLICATION EN 3840
8.1	GENERAL
8.2	SAFETY AND TRUE TO LIFE SITUATIONS
8.3	ELEMENTARY SET UP REQUIREMENTS
8.4	NECESSARY ACTIONS
8.5	MINIMUM NETWORK CONFIGURATIONS
8.6	ADDITIONAL SET-UP REQUIREMENTS
8.7	TESTABLE SITUATION
9	REQUIREMENTS FOR PRACTICAL LOCATIONS UNDER THE SCOP OF BEI BLS 35
10	REQUIREMENTS FOR PRACTICAL LOCATIONS UNDER THE SCOPE OF BEI BHS 35
10.1	GENERAL
10.2	SAFETY AND TRUE TO REALITY SITUATIONS
10.3	ELEMENTARY SET-UP REQUIREMENTS
10.4	SIMULATION WITH VIRTUAL REALITY
10.5	SPECIFIC SET-UP REQUIREMENTS
10.6	NECESSARY ACTIONS
10.7	MINIMUM NETWORK CONFIGURATIONS
10.8	ADDITIONAL SET-UP REQUIREMENTS
10.9	TESTABLE SITUATIONS
11	REQUIREMENTS FOR PRACTICAL LOCATIONS UNDER THE SCOPE OF VIAG. 38
	Image: 13 style="text-align: center;">
12	STRUCTURE OF MEETINGS
12.1	DELIBERATIVE BODIES
12.1. ⁻	1 BOARD OF DIRECTORS

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



12.1.2 STANDARDS MANAGEMENT COMMITTEE
12.1.3 EXPERT COMMITTEES 40
12.1.4 EXAMINATION BOARDS AND TASK FORCES
12.1.5 HARMONIZATION MEETINGS FOR EXAMINERS
12.1.6 HARMONIZATION MEETINGS FOR EXAMINERS
12.1.7 APPROVAL OF DOCUMENTATION
12.2 REQUIREMENTS CONCERNING THE STRUCTURE OF MEETINGS
12.2.1 COMPETENCIES
12.2.2 BALANCED COMPOSITION
13 CENTRAL SCHEME
13.1 HARMONIZED AND CENTRAL SCHEME
13.2 MAINTENANCE OF THE CENTRAL SCHEME
13.2.1 EVALUATION OF THE CENTRAL SCHEME
13.2.2 PROCESS FOR MAINTAINING THE CENTRAL SCHEME
14 PROFILE DOCUMENTS
14.1 CONTENT OF THE PROFILE DOCUMENTS
14.2 MAINTENANCE AND DEVELOPMENT OF NEW PROFILE DOCUMENTS
14.2.1 EVALUATION OF PROFILE DOCUMENTS AND ASSESSMENT REQUIREMENTS 44
14.2.2 PROCESS FOR MAINTAINING PROFILE DOCUMENTS
14.2.3 PROCESS FOR DEVELOPING NEW PROFILE DOCUMENTS
15 EXAM QUESTIONS
15.1 QUALITATIVE CRITERIA FOR THEORETICAL QUESTIONS
15.2 EVALUATION OF EXAM QUESTIONS
15.3 DUTIES OF THE EXAMINATION BOARD REGARDING THE DEVELOPMENT AND REPLACEMENT OF THEORETICAL QUESTIONS
15.4 PROCEDURE FOR DRAFTING AND REVISING THEORETICAL QUESTIONS 47

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



15.5 PRACTICAL ASSIGNMENTS NEN 3140 AND NEN 3840, BEI BLS AND BEI BHS 48
THE EXAMINATION BODIES MUST PREPARE AT LEAST FIVE PRACTICAL ASSIGNMENTS PER PROFILE
THE PRACTICAL ASSIGNMENTS MUST BE DEVELOPED AND MAINTAINED IN ACCORDANCE WITH THE APPLICABLE REGULATIONS:
15.6 PRACTICAL ASSIGNMENTS VIAG
PART 4: REQUIREMENTS AND INFORMATION FOR APPLICANTS, CANDIDATES AND CERTIFIED PERSONS
16 THE CERTIFICATION PROCESS
16.1 GENERAL DESCRIPTION OF THE CERTIFICATION PROCESS
16.2 APPLICATION PROCEDURE
16.3 DECISION ON CERTIFICATION
16.4 ISSUANCE OF CERTIFICATES
16.5 SUSPENSION OR REVOCATION
17. REGISTRATION OF CERTIFICATES
18 PROCEDURES FOR INSIGHT, COMPLAINTS, OBJECTIONS AND APPEALS 51
18.1.1 REQUEST FOR INSIGHT
18.1.2ASSESMENT OF THE REQUEST FOR INSIGHT
18.1.3THE FACILITATION OF INSIGHT
18.2 PROCEDURE FOR COMPLAINTS
18.2.1 GROUNDS FOR FILING A COMPLAIN
18.2.2 PROCEDURE FOR COMPLAINTS ABOUT THE STIPEL STANDARDS
18.2.3 PROCEDURE FOR COMPLAINTS ABOUT THE SERVICE PROVIDED BY STIPEL 53
18.2.4PROCEDURE FOR COMPLAINTS ABOUT THE SERVICE PROVIDED BY EXAMINATION BODIES
18.3 PROCEDURE FOR OBJECTIONS
18.3.1 GROUNDS FOR FILING AN OBJECTION
18.3.2 PROCEDURE FOR OBJECTIONS STIPEL

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



Introduction

This is a harmonized scheme to be used by all examination bodies that have an agreement to conduct examinations under the assessment requirements published by Stipel. This document also constitutes the overarching scheme for the assessment requirements specified by profile.

In view of the dangers of electrical work and operation work, Article 3.5 of the Working Conditions Decree stipulates that operations must be performed by competent and adequately trained workers. The guidelines for safe operation of electrical installations are described in NEN 3140 and NEN 3840 standards. These NEN standards constitute the national elaboration of the European standard NEN-EN 50110-1. Provisions of the NEN standards require no further reference.

In 1998, Stipel started certification of persons in accordance with ISO/IEC 17024 based on the regulations of NEN 3140 (low voltage) and NEN 3840 (high voltage). In 2018, about 10,000 person had a valid certificate for their work that includes work in energy production, industry, installation, utility construction and military defence. Certification is voluntary, but has a mandatory character. In a generally accepted way they provide the legal obligation for sufficient instruction regarding the safe execution of work on, in and with electrical installations.

In 2003, for the scope of public energy supply systems of the national grid operators, Netbeheer Nederland (Network Operation Netherlands) drafted its own regulations. In the so-called BEI BLS and BEI BHS, rules are set that apply specifically to work in grid management and these rules have additions to and deviations from NEN 3140 and NEN 3840. In addition, VIAG set rules for working on gas energy supply systems. Certification of persons in accordance with these regulations is required by all grid operators without any exceptions. In 2018, about 30,000 valid person certificates in the field of energy supply with electricity and gas were in circulation.

STIPEL has been personally doing the certification of persons in cooperation with examination bodies who are responsible for the examination process. The central scheme connects on a voluntary basis to the stipulations of ISO/IEC 17024:2012 eng.

The documents that contain the terms of assessment for the exams are defined by profile documens in the central scheme. STIPEL 10003:2020 contains the unified names for these profiel documents.

E. Weerepas Chairman Stipel

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



Editorial Remark of this document: This English version of STIPEL 10001:2025 is only applicable for NEN 3140 and NEN 3840 profiles. Therefore, all the text that relates to the BEI BLS, BEI BHS and VIAG profiles is non-authorative as stated when applicable.

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



Central scheme for conformity assessment of personen for handling safety hazards in power engineering

1 Subject and scope

The central scheme applies to all profiles of Stipel certification. The document has the following three related parts:

- Part 1: Requirements for the certification process
- Part 2: Requirements for the examination process
- Part 3: Requirements concerning the structure of meetings and documentation
- Part 4: Requirements and information for applicants, candidates and certificate holders

2 Normative references

The following documents must be applied in conjunction with this document.

Documents with a version number or publication date apply only in the edition mentioned. For documents without a version number or publication date, it must be assumed that only the latest published edition including any transitional arrangement and including any interim changes apply.

- ISO/IEC Guide 2:2004, Standardization and related activities General vocabulary
- NEN-EN-ISO/IEC 17000:2004 en, Conformity assessment Vocabulary and general principles
- NEN-EN-ISO/IEC 17024:2012 en, Conformity assessment General requirements for bodies performing certification of persons
- NEN-EN-ISO 9001:2015 nl, Quality management systems Requirements
- NEN 3140+A3:2019 nl, Operation of electrical installations Low voltage
- NEN 3840+A3:2019 nl, Operation of electrical installations High voltage
- BEI BHS, Operations of Electrical Installations Branch High Voltage (and Medium Voltage), Netbeheer Nederland (Network Operation Netherlands)
- BEI BLS, Operations of Electrical Installations Branch Low Voltage, Netbeheer Nederland (Network Operation Netherlands)
- VIAG, Natural gas safety instruction for power companies, Netbeheer Nederland (Network Operation Netherlands)
- VI Biogas, Biogas safety instruction for members of Netbeheer Nederland (Network Operation Netherlands)

Remark one of this paragraph: The NEN 3140 and NEN 3840 are to be revised every three years on average. The exam papers are continuously updated to reflect the state of the art and relevant standards. When a new version of these standards comes into effect, this document must be updated. The Board of Experts may make interim interpretation decisions, where necessary, which will modify the test terms to keep in line with the latest version of the standards. On 07-11-2019, the Board of Experts-Electrical Engineering made a decision with

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



number 38-2 and this decision reads "The textual consequences of the 2019 version have been discussed and processed by the CEC-E and the unanimous decision was taken to work within Stipel from 01-01-2020 with NEN 3140+A3:2019 and NEN 3840+A3:2019". All references in the STIPEL documents to NEN 3140 and NEN 3840 shall be deemed to refer to the version declared applicable in accordance with board decision 38-2.

Remark two of this paragraph: The BEI BHS, BEI BLS, VIAG and VI Biogas are reviewed annually and a new version comes into effect on April 15 each year. STIPEL's working groups, committees and colleges will conduct an annual impact analysis of proposed changes prior to publication. The examination papers are defined annually and, if necessary, the corresponding test requirements are adjusted.

3 Terms and definitions

This document uses the terms and definitions from:

- ISO/IEC Guide 2:2004
- NEN-EN-ISO/IEC 17000:2004 en
- NEN-EN-ISO/IEC 17024:2012 en

The terms and definitions where no source is given are specifically related to this document. For some other definitions, square brackets, ([and]), indicate the source from which the term and definition are derived and whether a modified definition is used. The modifications are related to the specific scope of this document and do not contain critical deviations from the commonly used terms and definitions.

3.1 appeal

request by an applicant, candidate or certificate holder for reconsideration of a decision taken by STIPEL regarding a complaint or objection [Source: ISO 17024 (own translation)]

3.2 applicant

person who has submitted an application to be admitted into the certification process and has signed the registration form in order to become a certified person [Source: ISO 17024 (own translation)]

3.3 assessment requirements

set of specific requirements relevant with a profile, including the central scheme requirements that must be met to become and remain certified [Source: ISO 17024 (own translation)]

3.4 body of experts

Body established by STIPEL that is responsible for developing, preparing and maintaining the profile documents on behalf of STIPEL

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



3.5 candidate

applicant who has fulfilled specified prerequisites and has been admitted to the certification process.

3.6 certified person

candidate who has passed the assessment and examination and received a STIPEL certificate [Source: ISO 17024 (own translation)]

3.7 certification process

activities by which STIPEL determines whether a person meets the assessment requirements including the application, assessment, certification decision, recertification and use of certification and the logo or certification mark [Source: ISO 17024 (own translation)]

3.8 competence

ability to apply knowledge and skills with an intended outcome [Source: ISO 17024 (own translation)]

3.9 complaint

request by an applicant, candidate or certificate holder to correct a deficiency/shortcoming in the service or assessment requirements

[Source: ISO 17024 (own translation)]

3.10 conformity assessment

process demonstrating a person's compliance with established requirements [Source: ISO 17000]

3.11 exam

mechanism that is part of an assessment in which the candidate's competence is determined according to the requirements of the central scheme in one or more ways, such as written, oral, observational or practical

[Source: ISO 17024 (own translation)]

3.12 examination body

An organization operating independently from any educational institution and contracted by STIPEL, entrusted with conducting examinations, developing and adapting practical exams, performing preliminary assessments of conducted practical exams, and providing documentation to enable STIPEL to make certification decisions.

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



3.13 examination requirements

set of documents for STIPEL to make clear and independent decisions whether a candidate has the competencies required in the specific requirements of the profile document and the central scheme

3.14 examiner

The person who takes and assesses an examination commissioned by an examination body, if the examination requires profession judgment [Source: ISO 17024 with changes]

3.15 impartiality

presence of objectivity [Source: ISO 17024 in our own translation]

Remark 1 with this term: Impartiality entails that no conflict of interest will arise or that these will be resolved, so that the activities in the certification process are not negatively affected.

Remark 2 with this term: Other terms which can be useful in describing impartiality are: independence, absence of conflicts of interest, absence of preferences, absence of prejudices, neutrality, fairness, broad-mindedness, distance and balance.

3.16 objection

request by an applicant, candidate or certificate holder for reconsideration of a decision made by STIPEL.

[Source: ISO 17024 (own translation)]

3.17 profile

Description of the subject on which the person or organization has competence

3.18 services

all activities performed by STIPEL as part of the certification process

3.19 secretary

A supporting body of STIPEL responsible for the administrative, organizational, and procedural support of the board, the other bodies within the governance structure, stakeholders, applicants, candidates, and certificate holders.

3.20 stakeholder

party who has an interest in the conformity assessment [Source: NTA 8813]

Remark 1 with this term: In this document the following parties are distinguished as stakeholders:

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



- a) the party who is identified as end user and as such makes use of the services of certificate holders (customers);
- b) the party identified as direct customer of the conformity assessment body (certificate holders and employers).

3.21 suspension

temporary invalidation of a certificate holder's STIPEL certification [Source: ISO 17024 (own translation)]

3.22 terms of assessment

required qualities listed in the profile documents regarding current knowledge, understanding, skills or professional attitudes of person to be certified

3.23 withdrawal

Termination of the certification of a certificate holder by STIPEL

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



Part 1: Requirements for the certification process

4 Content of the central scheme

4.1 General

The assessment requirements, specific interpretations or applications of this scheme may not conflict with ISO/IEC 17024 and legal provisions. They must not override the requirements of ISO/IEC 17011 or European Regulation (EC) 765/2008.

4.2 Purpose of the central schema

4.2.1 Operational safety

The purpose of the central scheme is to achieve occupational safety in power engineering by ensuring safe operations by competent, adequately trained and skilled persons. Safe operation means operations that function safely under normal and abnormal conditions according to the purpose for which they were designed.

Operations include all activities on or near energy installations, being electrical and power supply installations, to achieve the intended result:

- switching, operating, regulating and monitoring
- electrical, gas, energy and non-energy work such as construction, expansion, removal and maintenance of related parts and components
- taking and eliminating security measures
- inspection of the installation

Remark one of this paragraph: Safe operation aims to ensure the safe operation of an electrical system according to the purpose for which it was designed. Whether an electrical installation actually functions after operation is beyond the scope of the profile documents with assessment requirements. The scope includes only the recognition and prevention of hazards that may occur in the performance of operations activities.

4.2.2 Hazards

Hazards can occur when performing operating actions and performing work on or near energy technical installations (electrical or gas) and related work equipment. This risk may result in serious bodily injury due to:

- electric shock (electrocution)
- arc flashes
- explosion
- fire and other property damage
- intoxication
- electromagnetic forces
- unintentionally turning on and off or connecting and disconnecting

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



With regard to electrical installations, NEN 3140 and NEN 3840 set out the requirements and procedures to come to safe operations. With regard to electrical energy supply in the public grid operation in the Netherlands, these requirements are set by the BEI BLS and BEI BHS. With regard to gas energy supply in the public grid operation in the Netherlands, these requirements are set by VIAG.

The focus of the certificate for persons is on the prevention of the risks that persons run when carrying out work on electrical engineering installations as indicated in NEN 3140/NEN 3840, BEI BHS, BEI BLS, VIAG and related safety instructions.

A certificate for persons is the result of an independent review, and indicates that the certificate holder has demonstrated that he or she possesses the right knowledge, skills and competencies. Substantive knowledge of the discipline which is essential to be able to work safely will be reviewed to a limited degree. The certificate for persons is not, however, a substitute for a nomination as stated in the aforementioned regulations. This nomination – and the prescribed prior review – remains the responsibility of the employer.

Remark 1 with this paragraph: The application of these standards provides justified reliance, including on the part of the Netherlands Labour Authority, that hazards that arise are recognised and prevented and that occupational health and safety obligations in this area can be fulfilled. The occupational health and safety obligations refer with regard to electricity to, in particular, the Occupational Health and Safety Regulation, Art. 3.5(1) and (4) relating to electrical engineering, operating and other work on or near an electrical installation. The obligation entails that electrical engineering work and operating work which could entail hazards, must be carried out by skilled, adequately trained and authorised workers. The authorised worker must, in addition to effective measures, ensure the safe conducting of the work. With regard to working with gas, corresponding provisions have been included in the Occupational Health and Safety Regulation.

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



4.3 **Object of certification**

4.3.1 Elements to determine the scope

This scheme sets the requirements for the conformity assessment of persons who have or must have the competences for the safe operation of energy technical installations. The scope of conformity assessment is further prescribed in the assessment requirements.

The scope must be formulated according to the scope, powers and specializations with respect to the installation.

The areas of application are:

- installations under the scope of NEN 3140: low voltage electrical installations
- installations under the scope of NEN 3840: high voltage electrical installations
- installations under the scope of BEI BLS: public energy supply with low voltage electricity
- installations under the scope of BEI BHS: public energy supply with high voltage electricity, including medium voltage
- installations under the scope of VIAG: public energy supply with gas

The possible powers on the basis of NEN 3140 and NEN 3840 are:

- instructed person (VOP in Dutch)
- skilled person (VP in Dutch)
- control operator (BD in Dutch)
- nominated person in combined control of an electrical installation and work activity (IV/WV in Dutch)
- person involved in inspection (Inspection)

The possible powers on the basis of BEI BLS, BEI BHS and VIAG are:

- person having access (THP in Dutch)
- instructed person (VOP in Dutch)
- skilled person (VP in Dutch)
- allround skilled person (AVP in Dutch)
- nominated person in control of a work activity (WV in Dutch)
- control operator (BD in Dutch)
- nominated person in control of operating an electrical installation (OIV in Dutch)
- nominated person in control of an electrical installation (IV in Dutch)

The permitted profiles must be stated by name as shown in paragraph 4.3.3 of this scheme.

[Source: ISO 17024:2012 paragraph 8.2]

Remark one of this paragraph: Within the scope of NEN 3140 and NEN 3840, the competence of operations expert appears only in NEN 3840. The competence of team leader (TL) also

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



appears only in NEN 3840, but has no specific profile for certification. The person involved in inspection is not a competence mentioned in NEN 3140 or NEN 3840, but constitutes a specific profile in certification.

4.3.2 Duties and responsibilities

For the competences under the scope of NEN 3140 and NEN 3840, the following descriptions of responsibilities and duties are used:

- a nominated person in control of an electrical installation is a person with electrical engineering training who is responsible for ensuring at any time the safe status of the electrical installation or its components that are in use. The tasks which ensue therefrom are the drawing up of and seeing to the compliance with requirements and procedures relating to the safe execution of operating activities on the electrical installation or its components that are in use or are to be taken into use. The positions of the nominated person in control of an electrical installation and the nominated person in control of a work activity referred to hereafter can be held by one person
- a nominated person in control of a work activity is a person with electrical engineering training who is responsible for the safe progress of the execution of activities on the electrical installation. The tasks ensuing therefrom are making an inventory of the possible risks and the drawing up and ensuring of compliance with requirements and procedures to prevent those risks. Insofar as is necessary to prevent hazard it is necessary to temporarily turn off the electrical installation or its components, a procedure will be agreed in this respect with the nominated person in control of an electrical installation. A nominated person in control of a work activity must also be able to fulfil all tasks of the skilled person referred to hereafter
- a skilled person is a person with electrical engineering training who can be charged with the independent execution of all operating activities on instruction. In addition, he or she must be able to recognise possible hazards that might arise despite the requirements and procedures established by the nominated person in control of a work activity, and to prevent them by applying adequate safety measures
- an instructed person carries out specific work on or near electrical installations whereby electrical engineering hazards can arise. An instructed person is not an electrical engineer, but he or she is instructed to recognise the hazards that might arise in his or her work and to prevent them by applying adequate safety measures.

The authorisations and tasks under the scope of BEI BLS, BEI BHS and VIAG are defined in the aforementioned documents.

[Source: ISO 17024:2012 paragraph 8.2]

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



4.3.3 Authorised profiles

Certification is only allowed for the profiles established by STIPEL as listed in STIPEL 10003:2020 Uniform Designation of Profile Documents or a subsequently adopted newer version of this document.

[Source: ISO 17024:2012 par. 8.1]

4.4 Conformity assessment

The conformity assessment under this document consists of an assessment of prerequisites and a certification test

Assessment of the prerequisites may include:

- a theoretical examination in the areas of application under the scope of NEN 3140 and NEN 3840 consisting of a basic test unless the assessment requirements of the profile documents stipulate that this basic test is not required.
- (Non authorative text) a document review in the areas of application under the scope of BEI BLS, BEI BHS and VIAG consisting of a review of a pre-training requirement set forth in the relevant appendices of the BEI BLS, BEI BHS and VIAG. The conformity assessment in this case may only focus on the professional education and the required additional courses or training listed in the annexes and explicitly not on the experience requirements listed in the annexes.

The certification test may consist of:

- a theoretical examination, which in all cases, except for the certification of Inspection LV under the scope of NEN 3140 and NEN 3840, must come from STIPEL's central item bank;
- a practical exam, if required by the assessment of the profiles.

[Source: ISO 17024:2012 paragraph 8.2 and 8.3]

4.4.1 Conformity assessment NEN 3140 and NEN 3840

4.4.1.1 Prerequisites

For profiles on the NEN 3140/NEN 3840 scope, the candidate must pass the theoretical basic test associated with the profile:

- The result of the basic test has a period of validity of five years. After the expiration of this period, the candidate must conform to the above mentioned prerequisites.
- A candidate who fails the basic test may not retake the exam until 5 business days have passed, to prevent candidates from trying to pass the basic test by taking the test multiple times in a row and guessing the answers.

The entrance requirement does not apply to candidates who:

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



- Started a certification process for the profiles of instructed person (ILS).
- Are approved for recertification in accordance with paragraph 4.4.1.4.

Remark one of this paragraph: The basic test is an examination about general electrical engineering knowledge.

4.4.1.2 Certification test: theoretical examination

The candidate must take a theoretical exam in accordance to the requirements of paragraph 5.1 and the applicable profile document:

- The result of the theoretical part of the certification test for a profile under the scope of NEN 3140/NEN 3840 is valid for 1 year.
- A candidate who fails this theoretical exam may not retake the exam until 5 business days have passed, to prevent candidates from trying to pass this theoretical exam by taking the test multiple times in a row and guessing the answers.
- The time allowed for the theoretical examinations under the scope of NEN 3140 and NEN 3840 is defined in the assessment requirements of the applicable profile document.

Remark one of this paragraph: The duration of theoretical exams under the scope of NEN 3140 and NEN 3840 is currently 30, 45 or 60 minutes.

4.4.1.3 Certificatietest: practical examination

The candidate must take a practical examination according to the requirements of the applicable profile document:

- The practical exam consists of part A and part B depending on the requirements specified in the respective profiles.
- The practical examination consists of written practical assignments at practical locations that meet the requirements of chapter 6 under the scope of NEN 3140 and meets the requirements of chapter 7 under the scope of NEN 3840.
- The result of part A or B of the practical part of the certification test for a profile under the scope of NEN 3140/NEN 3840 is valid for 1 year.

4.4.1.4 Recertification

For the scope of NEN 3140 and NEN 3840, the candidate may be admitted to recertification if he holds a certificate for the same profile issued no more than five years ago.

The prerequisites of paragraph 4.4.1.1 may not be applied and the conformity assessment consists only of the requirements of the certification test, in the case of admission to recertification,

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



4.4.2 Conformity Assessment BEI BLS, BEI BHS and VIAG

4.4.2.1 Prerequisites

(Non authoritative text)

The candidate must meet prerequisite requirements for the profiles in the BEI BLS, BEI BHS and VIAG application areas:

- The profiles on the scope of the BEI BLS require compliance with the first and second columns of the training requirements as described in Annex 2 of the BEI BLS.
- The profiles on the scope of the BEI BHS require compliance with the first and second columns of the training requirements as described in Annex 3 of the BEI BHS.
- The profiles on the scope of the VIAG require compliance with the first and second columns of the training requirements as described in Annex 12 of the VIAG.

The entrance requirement does not apply to candidates who:

- Are approved for recertification in accordance with paragraph 4.4.2.4.

Remark one of this paragraph: The continuously updated Education and Experience Requirements are published by Netbeheer Nederland at www.beiviag.nl, which may include a reference to www.ckb.nl/opleidingstabel.

Remark two of this paragraph: The Board of Experts in its decision with number 87-3 dated 07-06-2018 determined that the possession of an SCC certificate, as required in a number of profiles, can not be tested by STIPEL. Possession of a VCA certificate is a requirement, which must be verified by the employer. Similarly, STIPEL may not test the work experience requirement. Work experience must be assessed by the employer as part of the designation. STIPEL must therefore test against 1) the prior education requirement and 2) the additional training requirement (second and third columns of the training tables).

4.4.2.2 Certification test: theoretical examination

(Non authoritative text)

The candidate must take a theoretical examination according to the requirements of paragraph 5.1 and the applicable profile document:

- The allowed time duration of all theoretical exams of scope BEI BLS, BEI BHS and VIAG is 60 minutes.
- The result of the theory part of the certification test on the scope of VIAG, BEI BHS and BEI BLS is valid for 6 months.
- A candidate who fails the theoretical exam may not retake the exam until after 5 business days to prevent candidates from trying to pass a theoretical exam by taking the exam multiple times in a row and guessing the answers.

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



4.4.2.3 Certification test: practical examination

(Non authoritative text)

The candidate must take a practical examination according to the requirements of the applicable profile document:

- The practical examination consists of a part A and B according to the requirements specified in the relevant profile document.
- The practical examination parts A and B consist of one or more written practical assignments at practical sites that meet the requirements of Chapter 8 of scope BEI BLS, meet the requirements of Chapter 9 of scope BEI BHS, or meet the requirements of Chapter 10 of scope VIAG.
- The result of parts A and B of the practical part of the certification test on the scope of VIAG, BEI BHS and BEI BLS is valid for 6 months.

4.4.2.4 Recertification

(Non authoritative text)

In the scope of BEI BLS, BEI BHS and VIAG, the candidate may be admitted to recertification if:

- they hold a certificate for the same profile issued not more than three years ago;
- the end of the certificate's validity period is no more than 6 months in the future.

In the case of admission to recertification, the prior education requirement of paragraph 4.4.2.1 may not be applied.

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



Part 2: Requirements for the examination process

5 General requirements theoretical and practical exams

5.1 General requirements theoretical exams

[Source: ISO 17024:2012 paragraph 9.3.1]

5.1.1 Conducting theoretical examinations

For all application areas and for all profiles, the theoretical exams is assed within the digital testing system TestVision, where the assessment of a completed test takes place automatically.

The presence of an examiner is not necessary during the theoretical examination. One supervisor without expertise of the assessed contents is sufficient.

5.1.2 Test form

A theoretical exam can consist of questions in one of the following question formats:

- the multiple-choice question: the candidate must choose the only correct answer from a number of answers;
- the true/false question: the candidate must indicate whether a statement is true or false;
- multi-select multiple choice: the candidate must choose several correct answers from a number of answers;
- the ranking question: the candidate must rank a number of details in the right order;
- the match question: the candidate must link a number of details to each other;
- the hotspot question: the candidate must point out the right answer on a photo or illustration;
- the fill-in question: the candidate must fill in a short answer in numbers or letters.

5.2 General requirements practical exams

- The practical exam is conducted under the responsibility of the Examination Body at exam locations that must comply with the requirements set forth in this document.
- When conducting a practical examination, such measures must be taken to ensure the safety of the candidate and other persons present at the examination
- A practical exam consists of completing one or more written practical assignments. The performance and outcome will be assessed using an assessment protocol associated with the assignment or assignments, also in writing.

[Source: ISO 17024:2012 paragraph 9.3.1]

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



5.3 Other general requirements exams

5.3.1 Language of the exam

Because certification is focused on the national market and communication is an important safety aspect, exams are conducted exclusively in the Dutch language.

Remark one of this paragraph: Language proficiency is an important safety aspect. That is why technicians must be able to speak, read and write Dutch at a sufficient level to ensure good communication. An appropriate level is B2 in accordance with the CEFR. The team members must be able to understand each other. Per team, at least one team member must be proficient in both Dutch and the language of the team (in case of non-Dutch speaking team members).

Remark two of this paragraph: For the scope of NEN 3140 and NEN 3840, it is now possible, as part of a pilot project, to take examinations in English and to issue certificates in English. Theoretical exams in English must be administered in TestVision.

5.3.2 Use of reference materials and dictionaries

The use of reference materials and dictionaries during the theoretical and practical exams is not permitted, unless explicitly allowed by STIPEL.

(Non-authoritative text)

- The use of reference materials in the form of the regulations for BEI BLS, BEI BHS, and VIAG is permitted during the respective exams.
- In such cases, the reference materials must be provided by the Examination Body in written or digital form to prevent candidates from using personal notes.
- If the use of reference materials brought by the candidate, in the form of the regulations for BEI BLS, BEI BHS, and VIAG, is permitted, the Examination Body must demonstrate that the reliability of the exam is not compromised thereby.
- The use of a general translating dictionary is allowed for individuals whose second language is Dutch.
- In such cases, the general dictionary must be provided by the Examination Body in written or digital form as far as reasonably possible, to prevent candidates from using personal notes.
- If the use of a general dictionary brought by the candidate is permitted, the Examination Body must ensure that the dictionary contains no personal annotations or explanatory notes.
- The use of a technical dictionary with explanatory notes or an encyclopedic work is explicitly prohibited.

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



Remark one of this paragraph: If an Examination Body exceptionally permits the use of a general dictionary based on its own examination regulations, it must verify that the candidate still possesses a sufficient level of proficiency in the Dutch language as specified in section 5.3.1.

5.3.3 Taking theoretical- and practical examination individually

All theory and practical exams must be completed individually by the candidate and without assistance from third parties. The following conditions must be ensured:

- Candidates are unable to copy from one another
- Candidates are not assessed based on interdependent roles
- Candidates are able to communicate with the examiner at all times (e.g. as sender/receiver), if such communication is part of the assignment
- Examiners can accurately observe all critical actions throughout the entire assignment (e.g. tests, measurements, etc.)
- Examiners are able to intervene or halt the exam at any time during the assignment and communicate this to the candidate
- The operation of the installation during the practical exam is not dependent on actions performed by another candidate

A practical exam involving multiple candidates may only be conducted simultaneously, in a uniform manner, or with role-based interdependence, if the examination institution has demonstrated based on a risk assessment and a specific action plan—that the above requirements are met.

5.3.4 Remote exams

- An examination body may conduct remote assessment examinations if the remote examinations meet the requirements of ISO 17024:2012 and IAF MD 4:2018 and the examination body has received a positive opinion from an ad hoc Task Force on Remote Examinations established by STIPEL.
- A Remote Examinations Task Force will be assembled by STIPEL on an ad hoc basis for the purpose of assessing an examination body's specific operating procedure, consisting of a representative from both the CoE Electrical Engineering and the CoE Network Management, with at least one of the representatives having taken or attended a remote pilot examination. Remote assessment may be considered for both recertification and initial certification.

Remote practical exams

- Notwithstanding NEN's message of March 18, 2020, the assessment of operational skills is not precluded from remote assessment provided that the assessment takes place in a physical practicum, where a supervisor is physically present and a remote examiner has full oversight of the practicum and the candidate's actions.

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



- The duration of a remote practical exam must be extended by 25% in view of the time that additional communication actions may require.

Remote theoretical exams

- A theoretical exam may be administered remotely provided that the exam is administered with the proctoring software, the identity of the candidate can be established, exam fraud is prevented, and the confidentiality of the exam questions is maintained.

Remark one of this paragraph: The remote examination scheme was introduced in 2020 for holding exams during the restrictive measures to prevent covid-19, but the remote examination scheme became permanent later that year.

Remark two of this paragraph: The proctoring software referred to relates specifically to Proctorio or ProctorExam as supported by Teelen in conjunction with TestVision.

5.4 Assessment of theoretical and practical exams NEN 3140 and NEN 3840

[Invulling van ISO 17024:2012 par. 9.3.1]

5.4.1 Assessment of theoretical and practical exams NEN 3140 and NEN 3840

- Assessment of the theoretical exams (basic and certification) are based on the test matrix established in TestVision. The test matrix, grade and duration of the theoretical exams (basic test and certification test) are shown in the applicable profile documents.
- The practical examinations are assessed by the examination bodies, using the assessment protocol as shown in STIPEL 20016:2022.

The theoretical and practical exams must be assessed separately. An examination for the profiles according to NEN 3140 and NEN 3840 will be satisfactory if:

- the basic test had a score of at least 70% of the maximum number of points that can be obtained;
- at least 70% of the maximum number of points that can be obtained was scored for the certification test of the theoretical examination;
- no more than 30 penalty points were given for the practical examination;

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



5.4.2 Assessment of theoretical and practical exams

- Assessment of the theoretical exams (basic and certification) are based on the test matrix established in TestVision. The test matrix, grade and duration of the theoretical exams (basic test and certification test) are shown in the applicable profile documents.
- The practical examinations are assessed by the examination bodies, using the assessment protocol as shown in STIPEL 20016:2022.

The theory and practical exams must be assessed separately. An examination for the profiles according to NEN 3140 and NEN 3840 will be satisfactory if:

- the basic theory test had a score of at least 70% of the maximum number of points to be obtained;
- at least 70% of the maximum number of points to be obtained was scored for the certification test of the theoretical examination;
- no more than 30 penalty points were given for the practical examination;

5.4.3 Assessment of theoretical and practical exams BEI BHS

(Non authoritative text)

- Assessment of the theoretical exams follow from the test matrix established in TestVision. The test matrix, grade and duration of the theoretical exams are shown in the applicable profile documents.
- The practical exams are assessed by the exam institutions, using assessment forms linked to the practical assignments.

The theory and practical exams must be assessed separately. An examination for the profiles according to BEI BLS will be satisfactory if:

- at least 70% of the maximum number of points to be obtained was scored for of the theoretical examination;
- no more than 30 penalty points were given for the practical examination;

5.4.4 Assessment of theoretical and practical examination VIAG

(Non authoritative text)

- Assessment of the theoretical exams follow from the test matrix established in Test-Vision. The test matrix, grade and duration of the theoretical exams are shown in the applicable profile documents.
- The practical exams are assessed by the exam institutions, using assessment forms linked to the practical assignments.

The theoretical and practical exams must be assessed separately. An examination for the profiles according to VIAG is considered satisfactory if:

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



- at least 70% of the maximum number of points to be obtained was scored for of the theoretical examination;
- no more than 30 penalty points were given for the practical examination;

6 General personnel requirements

6.1 Examiners

The examination bodies shall provide its examiners with documented and written instructions describing their duties and responsibilities.

The examiners that are responsible for conducting and assessing STIPEL examinations, must comply with the requirements of subject-matter expertise, knowledge of STIPEL-documentation and impartiality.

Subject-matter expertise for the assessment and evaluation of a practical exam must be of the following level:

- knowledge of the construction of the installation in which the test is to be taken
- at least 5 years' of experience with installations in the relevant area of application;
- knowledge of the standards and legislation, as stated in the relevant qualities to be tested;
- verifiable and up- to date knowledge on developments of the profession by working within the profession or engagement with the subject matter in any other verifiable capacity.

Knowledge of STIPEL-documentation:

- Having verifiable knowledge, insight and experience with the applicable profile document and central scheme so that the STIPEL-exams can uniformly be conducted and assessed.

Impartiality:

- The examiner must be verifiably impartial with regards to all parties that can take part in the certification process (applicants, candidates, certified persons);
- The examiner must be verifiably impartial with regards to candidates that have been prepared for the examination by a training institute at which the examiner has been involved with the training;
- Every official of the examination body must sign an agreement in which the confidentiality, impartiality and independence are guaranteed.

[Source: ISO 17024:2012 par. 5.2.3 en 6.2.2]

6.2 Persons who evaluate the exam results

The evaluation of the exam results – prior to the certification decision – must be carried out by an expert contracted by the examination body. He or she must at least possess the expertise required for examiners for the relevant exam in accordance with paragraph 6.1

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



7 Requirements for practical locations under the scope of NEN 3140

7.1 General information

A low voltage practicum must be suitable for the practical examination of persons that concern themselves with the operation and maintenance of low-voltage installations. These practical skills should involve tasks that may occur in low-voltage installations designed to closely resemble those used in actual practice.

7.2 Safety and true to life situations

Persons who execute practical tests in switching practice, must be able to execute these safely. At the same time, the installations must be set up in such way that they approximate practice as much as possible. This is the only way to acquire a reasonable degree of confidence that the candidate, after passing the test, will possess sufficient qualities to be able to perform work in low voltage installations.

The installations where the candidate carries out his or her test to obtain the certificate, must have been set up and maintained in such way that there can be no danger for all those present. Examining must take place under the management and supervision of skilled persons so that no hazardous situations can arise during the examination.

During the examination a nominated person in control of a work activity must be present who will ensure that no one is at risk. The nominated person in control of a work activity must be familiar with the installations to be able to take measures in the event of undesirable situations. The requirements relating to the set-up, equipment, installation and the actions to be carried out will be specified in further detail below.

Practical tests must be set up in such way that equality of testing is ensured.

Remark one of this paragraph: The installations which approximate the practice as best as possible will often be "utility-like" installations with the examination bodies, such as fuse units or sometimes a plastic distribution board (Halyester and the like). It will not be easy for an examination body to set up a metal-encased distribution system which is found in practice (such as Holec Capitole or ABB MNS). The requirement relating to approximation of the practice will therefore need to be described in further detail.

[Source: ISO 17024:2012 par. 7.4.1.]

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



7.3 Elementary set-up requirements

A low voltage practicum must be set up with low voltage components, realistic operating and control options, protective relay and other auxiliary equipment.

The earthing equipment, voltage testers, phase comparison, testing and selection equipment to be used must consist of approved devices which are actually applied in practice. The exams are carried out under operating voltage of 400/230 VAC.

The following safety requirements are set for the installations:

- 1. all safety measures must be taken as these apply for operating low voltage installations.
- 2. an up to date, clear and orderly technical file concerning the installations is available, which includes, in addition, the risk analysis of the installation. This technical file is available for inspection by the parties responsible for examining and certification.
- 3. the installations must be maintained in a skilled manner.

Requirements relating to the equipment and installation:

- Diversity in types of switching material (power switches, switch disconnectors, isolators and contactors).
- Diversity of brands of switch material.

[Source: ISO 17024:2012 par. 7.4.2]

7.4 Available materials

At least the following must be available for practical exams in the area of application of NEN 3140:

- all necessary measuring tools, such as: bipolar tracer (Duspol), multimeter, clamp-on ammeter, megger and installation tester (earth leakage and impedance measurement).
- personal protective equipment, like: rubber safety gloves, rubber safety mat, hard hat with face shield, blade fuse puller with leather cuff, standard blade fuse puller and cotton (not easily flammable) dust coat.
- all necessary auxiliary materials, like: earthing equipment, warning signs and/or labels, locks and other lock-out and cut-off tools.
- an inventory must be available of these materials, with any particulars.

[Source: ISO 17024:2012 par. 9.3.4]

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



7.5 Necessary actions

It must be possible to test candidates for the following aspects:

- Replacing lamps.
- De-energising before work and starting up again after work on electrical motors.
- Being able to connect and disconnect electrical motors.
- Fitting plugs to wiring.
- Replacing wall sockets and light switches.
- Connecting lighting fixtures to existing wiring.
- Inspecting electrical work tools.
- Putting in and removing fuses.
- Commissioning and decommissioning of motors, machines, final circuits, distribution boards, switchboards, lighting circuits, power supply cables, transformers and other installation components.
- Switching on, switching off and isolating and separating in a way that can be monitored, and if necessary earthing and short-circuiting of motors, machines, final circuits, distribution boards, switchboards, lighting circuits, power supply cables, transformers and other installation components.
- Being able to handle switchgear, switching conditions and procedures.
- Being able to apply safety regulations relating to switching, testing, earthing,
- warning and locking.
- Phase comparison and determining the phase sequence.
- Materials for communication if the situation requires this.

[Source: ISO 17024:2012 par. 7.4.2]

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



8 Requirements for practical locations for the are of application of NEN 3840

8.1 General

A high voltage practicum must be suitable for the practical examination of persons that concern themselves with the operation and maintenance of high-voltage installations. These practical skills should involve tasks that may occur in high-voltage installations designed to closely resemble those used in actual practice.

8.2 Safety and true to life situations

Persons who execute practical tests in switching practice, must be able to execute these safely. The installations where the candidate carries out the test to obtain the Stipel certificate, must have been set up and maintained in such way that there can be no danger for anyone present.

The following safety requirements are set for the installations:

- All safety measures must be taken as these apply for operating high voltage installations.
- A clear and orderly technical file is available for the installations. A risk analysis must be drawn up for the use. This technical file is available for inspection by the parties responsible for examining and certification.
- A risk analysis will be drawn up again for changes in installations. These changes will be recorded in the technical file.
- The installations must be maintained in a skilled manner.

Examining must take place under the management and supervision of skilled persons so that no dangerous situations can arise during the examination.

During the examination a nominated person in control of a work activity must be present who will ensure that no one is at risk. The nominated person in control of a work activity must be familiar with the installations to be able to take measures in the event of undesirable situations. The requirements relating to the set-up, equipment, installation and the actions to be carried out will be specified in further detail below.

At the same time, the installations must be set up in such way that they approximate practice as much as possible. This is the only way to acquire a reasonable degree of confidence that the candidate, after passing the test, will possess sufficient qualities to be able to perform work in high voltage installations.

Practical testing must be set up in such way that equality of testing is ensured.

[Source: ISO 17024:2012 par. 7.4.1.]

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



8.3 Elementary set up requirements

A high voltage practicum must be set up with high and low voltage components, realistic control and operating options, protective relay and other auxiliary equipment.

The earthing equipment, voltage testers, phase comparison, testing and selection equipment to be applied must consist of approved devices which are actually used in practice.

The operating voltage of the installations does not have to be high voltage.

[Source: ISO 17024:2012 par. 7.4.2]

8.4 Necessary actions

It must be possible to carry out the following actions:

- Commissioning and decommissioning a cable section.
- Commissioning and decommissioning transformers.
- Commissioning and decommissioning a high voltage station.
- Changing transformer or rail by means of remote control.

[Source: ISO 17024:2012 par. 7.4.2]

8.5 Minimum network configurations

It must be possible to set up the network in which switching work is to be carried out as a:

- Star network.
- Ring network.
- Meshed network.
- Sufficient options for through-connection at low voltage level.
- Realistic set-up of position indicators, lock-outs, signals, operating and control options.
- Availability of a control panel for the execution of remote operating and control actions.
- Sufficient options relating to dealing with malfunction indicators, protective relay.
- Interpreting and resetting malfunction signals and other auxiliary equipment.
- The entire set-up must comply with the applicable legislation and regulations.

[Source: ISO 17024:2012 par. 7.4.2]

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



8.6 Additional set-up requirements

The installation must possess:

- Diversity of types of switching material (power switches, switch disconnector, isolators).
- Diversity of brands of switching material HV and LV.
- Installations of the type "Open", "Encased" and "Closed". Diversity of protection facilities, including the absence thereof, at the start of a test.
- All necessary tools and personal protective equipment must be present, such as earthing equipment, voltage testers, phase comparison equipment, testing equipment, selection and colour-coding equipment, warning signs, locks and other lock-out and cut-off tools.
- An inventory must be available of these materials, with any particulars.

[Source: ISO 17024:2012 par. 7.4.2]

8.7 Testable situation

It must be possible to test candidates for the following aspects:

- Switching on, switching off and isolating and separating in a way that can be monitored of cables, transformers and other installation and network components.
- Resetting to the network configuration for normal business operations.
- Commissioning and decommissioning of a high voltage station or installation.
- Parallel connection of transformers and the changing of transformers.
- Changing, isolating, separating in a manner that can be monitored and earthing of rails.
- Being able to handle switchgear, switching conditions and procedures.
- Earthing network and installation components.
- Being able to apply safety regulations relating to switching, testing, earthing, warning and locking.
- Being able to handle voltage testers, earthing equipment, testing equipment.
- Phase comparison and determining the phase sequence.
- Interpreting and handling short-circuit indicators and safety signals
- Selecting a cable in a cable duct.
- Giving a cable a colour code.
- Testing a cable.
- Executing the correct measuring and test procedure when earthing network and installation components, opening ring/closing ring, connecting/disconnecting, through-connection/ending through-connection.
- Materials for communication if the situation requires this.

[Source: ISO 17024:2012 par. 8.3]

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



9 Requirements for practical locations under the scop of BEI BLS

(Non authoritative text)

No specific requirements apply to the practical test locations for the area of application of BEI BLS. The requirements for the practical test locations ensue from the practical assignments which are established by the BEI BLS work group and are approved by the Panel of Experts for Grid Operations.

10 Requirements for practical locations under the scope of BEI BHS

10.1 General

(Non authoritative text)

With a high voltage practicum, practical tests are to be taken relating to operating and securing high voltage installations. With these practical skills the matter must concern actions which can occur in commissioned high voltage installations as these are applied in the installations of transport and distribution companies.

10.2 Safety and true to reality situations

(Non authoritative text)

Persons who execute practical tests in switching practice must be able to execute these safely. At the same time, the installations must be set up in such way that they approximate practice as much as possible. This is the only way to acquire a reasonable degree of confidence that the candidate, after passing the test, will possess sufficient qualities to be able to perform work in high voltage installations.

The installations where the candidate carries out his or her test to obtain the certificate, must have been set up and maintained in such way that there can be no danger for the persons present. Examining must take place under the leadership and uninterrupted supervision of skilled persons so that no dangerous situations can arise during the examination. Practical tests must be set up in such way that equality of testing is ensured.

During the examination someone (e.g. examiner or supervisor) must be present who will ensure that no one is at risk. This person must be familiar with the installations to be able to take measures in the event of undesirable situations. He or she must have a nomination appropriate to his or her tasks, to be determined by the location manager. Safety during examining falls under the responsibility of a nominated person in control of a work activity. The nominated person in control of a work activity does not have to be present at the exam location at all times.

[Source: ISO 17024:2012 par. 7.4.1]

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



10.3 Elementary set-up requirements

(Non authoritative text)

A high voltage practicum must be set up with high voltage components, low voltage components (for the domains Transport and Medium Voltage), realistic operating and control options, protective relay and other auxiliary equipment. The earthing equipment, voltage testers, phase comparison, testing and selection equipment to be used must consist of approved devices which are actually applied in practice. The operating voltage of the installations does not require a high voltage level.

The set-up, equipment, installation of the practical exam location and the actions to be carried out must satisfy relevant technical standards and regulations.

[Source: ISO 17024:2012 par. 9.3.1]

10.4 Simulation with virtual reality

(Non authoritative text)

For the execution of practical assignments for certificates in the sub-domain Transport, with the prior consent of the Panel of Experts for Grid Operations, use can also be made of virtual high voltage installations, whereby the operating actions and safety actions are simulated by virtual reality. The candidate must then be able to name what operational situation he or she should find or will realise, how he or she will provide a safe workplace or can assess that a safe workplace has been provided and what measuring and testing equipment and earthing equipment are necessary. The components, equipment, tools and warning signs must actually be pointed out in the installation/the field.

10.5 Specific set-up requirements

(Non authoritative text)

The following safety requirements are set for the installations:

- All safety measures must be taken as these apply for commissioned installations.
- A clear and orderly technical file is available for the installations. A risk analysis must be drawn up for the use. This technical file is available for inspection by the parties responsible for examining and certification.
- A risk analysis will be drawn up again after changes in installations. These changes will be recorded in the technical file.
- The installations must be maintained in a skilled manner.

[Source: ISO 17024:2012 par. 7.4.2]

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



10.6 Necessary actions

(Non authoritative text)

It must be possible to carry out the following actions:

- Commissioning and decommissioning of a cable section.
- Commissioning and decommissioning of transformers.
- Commissioning and decommissioning of an MV space (not for the domain TR).
- Changing transformer or rail by means of remote control (for the domain TR).

[Source: ISO 17024:2012 par. 7.4.2]

10.7 Minimum network configurations

(Non authoritative text)

It must be possible to set up the network in which switching work is to be carried out like a:

- Star network. (only for MV, not for Transport)
- Ring network.
- Meshed network (only for MV, not for Transport)
- Sufficient options relating to through-connection at low voltage level (for the domains TR and MV).
- Realistic set-up of position indicators, lock-outs, signals, operating and control options.
- Availability of a control penal for the execution of remote operating and control actions. (only TR)
- Sufficient options relating to handling malfunction indicators, protective relay, interpreting and resetting of malfunction signals and other auxiliary equipment.
- The entire set-up must comply with the applicable legislation and regulations.

[Source: ISO 17024:2012 par. 7.4.2]

10.8 Additional set-up requirements

(Non authoritative text)

Additional requirements relating to the equipment and installation:

- Diversity of types of switching material (power switches, switch disconnector, isolators).
- Diversity of brands of switching material HV.
- Installations of the type "Open", "Encased" and "Closed" for voltages smaller than 110 kV
- Installations of the type "Open" and "Closed" for voltages greater than or equal to 110 kV
- All necessary tools and personal protective equipment must be available, such as earthing equipment, voltage testers, phase comparison equipment, testing equipment, selection and colour-coding equipment, warning signs, locks and other lock-out and cut-off tools.
- An inventory must be available of these materials, with any particulars.

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



[Source: ISO 17024:2012 par. 7.4.2]

10.9 Testable situations

(Non authoritative text)

It must be possible to test candidates for the following aspects:

- Switching on, switching off and isolating cables, transformers and other installation and network components.
- Resetting to the network configuration for normal business operations.
- Commissioning and decommissioning of a high voltage station (only MV) or installation
- Parallel connection of transformers and the changing of transformers.
- Changing, isolating, separating in a manner that can be monitored and earthing of rails.
- Being able to handle switchgear, switching conditions and procedures.
- Earthing network and installation components.
- Being able to apply safety regulations relating to switching, testing, earthing, warning and locking.
- Being able to handle voltage testers, earthing equipment, testing equipment.
- Phase comparison and determining the phase sequence.
- Interpreting and handling short-circuit indicators and safety signals.
- Selecting a cable in a cable bed.
- Testing a cable.
- Executing the correct measuring and test procedure when earthing network and installation components, ring-opening/ring-closing, connecting/disconnecting, through-connection/ending through-connection.
- Materials for communication if the situation requires this.

[Source: ISO 17024:2012 par. 8.3]

11 Requirements for practical locations under the scope of VIAG

(Non authoritative text)

No specific requirements apply to the practical test locations for the area of application of VIAG. The requirements for the practical test locations ensue from the practical assignments which are established by the VIAG work group and are approved by the Panel of Experts for Grid Operations.

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



Part 3: Requirements concerning the structure of meetings and documentation

12 Structure of meetings

12.1 Deliberative bodies

[Source: ISO 17024:2012 par. 5.1.2]

12.1.1 Board of Directors

- STIPEL shall be governed by a Board of Directors composed of a number of natural persons equal to the number of stakeholder groups, with each group appointing one board member.
- If the number of participating stakeholder groups is fewer than five, each group may appoint one or more additional board members, provided that each group appoints an equal number of board members.
- The Board must appoint an independent chairperson.
- The composition, responsibilities, and operation of the Board shall be further defined in the Articles of Association.

[Source: Articles of Association]

12.1.2 Standards Management Committee

- The Board of directors shall establish a Standards Management Committee (SMC) and appoint its members.
- The composition of the committee must ensure both the committee's expertise and the adequate representation of stakeholders.
- The chairpersons of the Expert Committees shall, where possible, be appointed in the Standards Management Committee.

The tasks of the SMC include:

- drafting the central standard in accordance with Chapter 13;
- providing solicited and unsolicited advice to the Board concerning the admission of stakeholders, the composition of the Board, the appointment of Board members, the assessment requirements for specific profiles, and the interpretation and consistent application of the central standard;
- handling appeals as referred to in paragraphs 18.4.2 and 18.4.3.

The composition, responsibilities, and operation of the Committee shall be further defined in the Articles of Association.

[Source: Articles of Association]

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



12.1.3 Expert Committees

The Board of Directors shall, upon request from a specific sector or industry, establish an Expert Committee (EC) for defined areas of expertise.

The tasks of the Expert Committee include:

- developing, drafting, and maintaining the assessment criteria for specific profile documents and examination requirements within their scope, in accordance with Chapter 14;
- discussing aggregated and anonymized results of theoretical examinations, including the pass rates of such exams;
- handling appeals as referred to in paragraph 18.4.2.

The Board of Directors has established two Expert Committees:

1) Expert Committee for Electrical Engineering (EC-EE), responsible for the profile documents, assessment criteria, and examination requirements within the scope of NEN 3140 and NEN 3840.

2) Expert Committee for Grid Management (ECGM), responsible for the profile documents, assessment criteria, and examination requirements within the scope of BEI BLS, BEI BHS, and VIAG.

The composition, responsibilities, and operation of the Committees shall be further defined in the Articles of Association.

[Source: Articles of Association]

12.1.4 Examination boards and task forces

The expert committees may establish examination boards and task forces.

An examination board is responsible for:

- Developing, drafting, and maintaining the questions for theoretical and practical exams within their scope of application in accordance with Chapter 15;
- Continuous evaluation of the scoring percentages of specific theoretical questions;
- Handling specific questions and remarks from stakeholders, explicitly including questions and remarks from certificate holders.

The Expert Committee for Network Management has established two examination boards: 1) The BEI Examination Board (EB-BEI), responsible for drafting exam questions under the operation of BEI BLS and BEI BHS.

2) The VIAG Examination Board (EB-VIAG), responsible for drafting exam questions under the operation of VIAG.

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



The Expert Committee for Electrical Engineering has established two examination boards:

1) The Central Electrical Engineering Examination Board (CEC-E), responsible for drafting exam questions under the operation of NEN 3140 and NEN 3840.

2) The Technical Committee STIPEL (TC-English), responsible for drafting exam questions under the operation of NEN 3140 and NEN 3840 within the framework of the STIPEL English pilot project.

[Source: Articles of Association]

12.1.5 Harmonization meetings for Examiners

The STIPEL Board shall establish a harmonization meeting for examiners in which all examination bodies authorized to conduct examinations under a cooperation agreement with STIPEL are represented.

- Each examination body shall be represented in this meeting by at least one lead examiner.
- The harmonization meeting is responsible for standardizing the implementation of examinations by the examination bodies.

12.1.6 Harmonization meetings for Examiners

The STIPEL Board shall establish a consultation meeting for training providers who offer courses/trainings/education programs concluded with a STIPEL examination and who have entered into a cooperation agreement with STIPEL.

The meeting is tasked with improving alignment with the market, enabling training providers to offer courses with sufficient frequency, so that employees in the energy technology sector are able to prepare for the examination corresponding to all profiles possible under the certification scheme.

12.1.7 Approval of documentation

- The documentation of the central certification scheme must be approved by the Board. The Board may seek advice on the drafting, evaluation, and amendment of this documentation from the Scheme Management Committee.
- The required documentation other than the central certification scheme must be drafted and approved by the competent expert committee.
- All documents, regardless of whether they are drafted by the secretary or the competent bodies of STIPEL, shall be approved for suitability for publication by the secretary on behalf of the Board.

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



12.2 Requirements concerning the structure of meetings

12.2.1 Competencies

All members of the meeting structure must declare in the statement attached as Annex A in which of the following fields they possess expertise:

- Electrical engineering
- Low-voltage energy supply
- High-voltage energy supply
- Gas energy supply
- Certification and examination
- Other specific fields of expertise to be named

The representatives of stakeholders in the Scheme Management Committee must collectively possess the expertise necessary to develop and maintain the central scheme.

Members of an examination board must collectively have:

- At least 5 years of experience in creating and assessing theoretical and practical examinations; and
- Subject matter knowledge in the relevant profile documents with assessment requirements, including:
 - At least 5 years of experience with high- and low-voltage installations;
 - A completed education at level 4 according to the NLQF;
 - Knowledge of standards and legislation, as specified in the applicable examination criteria.

[Source: ISO 17024:2012 par. 6.1.3]

12.2.2 Balanced composition

STIPEL shall ensure active and balanced representation of stakeholder parties, whereby representation of at least the following parties must be realized:

- The party identified as the end user, who makes use of the services of certificate holders (clients);
- The party identified as the direct customer of the conformity assessment body (certificate holders and employers).
- Stakeholder parties shall reasonably be entitled to representation in the consultative bodies of STIPEL.
- When appointing representatives of stakeholders to the meeting structure, the Board shall record on behalf of which organizations or umbrella organizations of stakeholder parties, such as industry or professional associations, the individual participates.

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



13 Central Scheme

13.1 Harmonized and central scheme

The central scheme is harmonized in the sense that STIPEL and every examination body must adhere to this scheme and are not permitted to deviate from it. The scheme is a central scheme in that it applies to all STIPEL profile documents.

13.2 Maintenance of the central scheme

13.2.1 Evaluation of the central scheme

The operation and implementation of the central scheme must be evaluated at least every five years by the Scheme Management Committee.

- The evaluation shall take place during a duly announced meeting.
- The Scheme Management Committee and all stakeholders shall be enabled to submit written comments regarding the evaluation in advance.
- All stakeholder parties may raise specific topics concerning the operation and implementation of the central scheme at any time for discussion by the Scheme Management Committee.
- These topics must be scheduled on the agenda and discussed, and where necessary, the Scheme Management Committee may take interim decisions.

13.2.2 Process for Maintaining the central Scheme

The maintenance of the central scheme shall include the following steps:

- 1) Evaluation by the Scheme Management Committee as described in paragraph 13.2.1;
- 2) Inventory of market parties that may have an interest in the proposed changes, including:
 - a. Certificate holders' clients;
 - b. Clients commissioning work from certificate holders;
- 3) Elaboration of the changes by or on behalf of the Scheme Management Committee;
- 4) Submission of the amended central scheme to the Board;
- 5) Approval and ratification of the central scheme by the Board;
- 6) Determination of the effective date for the use of the central scheme.

[Source: ISO 17024:2012 par. 8.5, 9.2 en 9.3]

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



14 Profile documents

14.1 Content of the Profile Documents

A profile document must at least consist of the following components:

- Scope of application
- Competency and job profile
- Examination criteria
- Assessment criteria, consisting of:
 - Assessment method;
 - Assessment matrix;
 - Mode of evaluation;
 - Standards and cut-off scores.

[Source ISO 17024:2012 par. 8.2]

14.2 Maintenance and Development of New Profile Documents

14.2.1 Evaluation of Profile Documents and Assessment Requirements

The operation and implementation of a profile document must be evaluated every five years by the competent expert committee:

- The evaluation shall take place during a duly announced meeting of the expert committee, and all stakeholders shall be enabled to submit written comments regarding the evaluation in advance.

During the evaluation of a profile document, the following must be demonstrated or confirmed:

- Stakeholder parties and support;
- Purpose or intended outcome of the profile document;
- Important and critical tasks performed by certificate holders;
- Certification requirements and associated assessment methods;
- If an oral or written exam is part of the assessment, an overview of the content:
 - Type of questions to be asked;
 - Knowledge level of the questions;
 - Number of questions for each topic;
 - Duration of the examination;
 - Method of evaluation.

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



The operation and implementation of the assessment requirements must:

- For the scope falling under BEI BLS, BEI BHS, and VIAG, be evaluated annually on an interim basis by the Expert Committee for Grid Management, in response to the annual updates of these regulations.
- For the scope falling under NEN 3140 and NEN 3840, be evaluated on an interim basis by the Expert Committee for Electrical Engineering whenever these standards are revised.

[Source: ISO 17024:2012 par. 7.2.2]

Remark one of this paragraph: See remark two under paragraph 2. With regard to BEI BLS, BEI BHS, and VIAG, proposed amendments are announced in October of the calendar year by Netbeheer Nederland. The Expert Committee for Grid Management determines the impact of the amendments on the assessment requirements and exam questions and proposes the necessary changes so that decisions can be made in time for the changes to take effect on 15 April of the following calendar year, simultaneously with the implementation of updates to BEI BLS, BEI BHS, and VIAG.

14.2.2 Process for maintaining profile documents

The maintenance of a profile document must include the following steps:

- 1) Evaluation of the profile document by the competent expert committee, as described in paragraph 14.2.1;
- 2) Decision by the expert committee, based on the evaluation, as to whether changes are necessary;
- 3) Elaboration of the changes by or on behalf of the expert committee;
- 4) Approval of the revised profile document by the expert committee;
- 5) Determination of the effective date for the use of the revised profile document.

[Source: ISO 17024:2012 par. 8.5]

14.2.3 Process for developing new profile documents

The development of a new profile document must include the following steps:

- 1) Assessment by the competent expert committee as to whether the intended profile document aligns with the objectives of STIPEL;
- 2) Identification of market parties that may have an interest in the outcomes of the conformity assessment activity:
 - Certificate recipients;
 - Clients commissioning work from certificate holders;
- 3) Verification of stakeholder support by establishing that multiple parties are actively willing to contribute to the development and maintenance of the scheme;
- 4) Description of the specific domain to which the profile applies;

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



- 5) Description of the requirements concerning competencies, assessment, and procedures, including requirements for any supervision and recertification;
- 6) Validation that the profile document contains the components described in paragraph 14.1;
- 7) Execution and evaluation of a pilot to determine whether the profile document is feasible and valid;
- 8) Approval and ratification of the profile document by the expert committee;
- 9) Determination of the effective date for the use of the scheme.

[Source: ISO 17024:2012 par. 8.5]

15 Exam questions

15.1 Qualitative criteria for theoretical questions

Questions must meet the following criteria:

- Relevant: A question must assess the knowledge as intended in the corresponding assessment term, both in content and level.
- Objective:

The indicated correct answer must be entirely accurate; the two distractors must be incorrect.

- Efficient:

The question stem must not contain any information that is unnecessary for answering the question correctly. The phrasing must ensure that only the intended knowledge is assessed.

- Specific:

-

The stem must not contain any clues pointing toward the correct answer or distracting from it. A correct answer should only be possible if the intended knowledge is mastered. Language use and level:

Language proficiency must not be a factor in the ability to answer the question correctly.

15.2 Evaluation of exam questions

The evaluation of exam questions must take place on an ongoing basis. Evaluation may be initiated based on:

- Questions and comments from candidates;
- Changes in the state of the art and applicable regulations;
- Analysis of incidents and accidents.

Evaluation must be carried out based on:

- Changes in relevant regulations;
- Questions, comments, and complaints from candidates;
- Annual analysis of questions with exceptionally high or low scores, based on a p-value lower than 0.15 or higher than 0.90;
- Annual analysis of distractors based on an a-value lower than 0.05 or higher than 0.30.

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



[Source: ISO 17024:2012 par. 8.5]

15.3 Duties of the Examination Board Regarding the Development and Replacement of Theoretical Questions

An Examination Board must:

- Determine, based on the evaluations outlined in paragraph 15.2, whether questions need to be revised;
- Ensure that for each profile document with assessment criteria, a sufficient number of questions is available to compile three entirely distinct exams (100% different) that comply with the test matrix of the respective profile document;
- Achieve a refresh rate of 5% of theoretical questions within their scope every three years, through the addition of new or revision of existing theoretical questions.

15.4 Procedure for Drafting and Revising Theoretical Questions

When drafting or revising exam questions, the following guidelines must be observed:

- Determine for which profile the question is intended.
- Determine the relevant assessment criterion (toetsterm) and identify the knowledge to be assessed.
- Formulate the question stem based on the knowledge to be assessed:
 - The stem may contain only one question.
 - It must be concise and free of unnecessary or misleading information.
 - It must be unambiguous in interpretation.
- Formulate the correct answer and the distractors:
 - The correct answer must be entirely accurate.
 - The distractors must be clearly incorrect.
 - Do not use alternatives such as "both answers are correct."
 - The distractors must align with the language used in the stem.
 - Distractors must not be nonsensical; they must relate to the content of the stem.
- Regarding language use and complexity:
 - Use clear and simple wording.
 - Use short sentences; divide longer texts into multiple sentences.
 - Preferably formulate the stem in a positive form.
 - Avoid double negatives in the stem.
 - The length of the correct answer and the distractors should be as equal as possible.

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



15.5 Practical assignments NEN 3140 and NEN 3840, BEI BLS and BEI BHS

The examination bodies must prepare at least five practical assignments per profile.

The practical assignments must be developed and maintained in accordance with the applicable regulations:

- The applicable regulations for drafting practical assignments under the scope of NEN 3140/NEN 3840 are set out in STIPEL 20016:2022 Development and assessment of practical assignments (NEN 3140/NEN 3840).
- The applicable regulations for drafting practical assignments under the scope of BEI BLS are set out in STIPEL 30016:2025 Development and assessment of practical assignments (BEI BLS).
- The applicable regulations for drafting practical assignments under the scope of BEI BHS are set out in STIPEL 40016:2021b Development and assessment of practical assignments (BEI BHS).

15.6 Practical assignments VIAG

The practical assignments for the scopes of VIAG are developed and prescribed by the competent examination committees and the College of Experts - Network Management. STIPEL is responsible for drafting an assessment protocol for each practical assignment.

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



Part 4: Requirements and information for applicants, candidates and certified persons

16 The certification process

16.1 General description of the certification process

Phase 1: Application

- 1) Applicant applies at an examination body;
- 2) applicant completes the digital registration form (annex A) either individually or in cooperation with the examination body;
- 3) Applicant signs the registration form and becomes a candidate.

Phase 2: Entrance (If there is a recertification in accordance with paragraph 4.4 of this document, steps 4a and 4b do not apply to the certification process):

4a) if the applicant has applied for a BEI BLS, BEI BHS or a VIAG certification procedure, an examination body shall conduct a screening process consisting of a check of prior education requirements in accordance with the training tables.

4b) if the applicant has applied for a NEN 3140 or a NEN 3840, during the certification process a basic test must be performed in accordance with paragraph 4.4.1;

Phase 3: Examination

- 5) a candidate is called up for the examination by the examination body, the scope of which depends on the applicable profile document;
- 6) a candidate performs the required examinations.
- Phase 4: Assessment
 - 7) the examination body enters the examination results into the STIPEL application which proves compliance with the examination requirements;
 - 8) the STIPEL application performs an automated evaluation based on the information collected by the examination body. STIPEL takes a positive decision on certification ff the certification criteria are met.

Phase 5: Issuance of the certificate

- 9) STIPEL issues a digital certificate with a specified period of validity to the candidate by sending this certificate to the email address provided by the candidate in the registration form;
- 10) The candidate is now a certified person and must (continue) to fulfil the obligations listed in the registration form and Terms & Conditions (see Annex 1 and 2)

[Source: ISO 17024:2012 par. 9.1.1]

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



16.2 Application procedure

An applicant may apply for certification under this central certification scheme by contacting an examination body that is authorized to examine under a cooperative agreement with STIPEL. To find these exam institutions go to https://stipel.nl/direct-regelen/examenbureau-vinden/.

- Next, the registration form in Annex A must be filled in and signed by the applicant.
- If the examination body is in possession of the applicant's e-mail address, this registration form can be filled in individually by the applicant.
- If this e-mail address is not present, an applicant must complete the registration form in cooperation with the examination body.

[Source: ISO 17024:2012 par. 9.1.2]

Remark one of this paragraph: The registration form of Annex A is subject to the applicable Terms & Conditions of Annex B.

16.3 Decision on certification

Decisions on certification are made by verifying that candidates have met the certification requirements of:

- this central certification scheme;
- the applicable profile documents.

The evidence of compliance with these certification requirements are collected exclusively during the certification process by the examination body(s), where the candidate has applied and taken the examination.

[Source: ISO 17024:2012 par. 9.4]

16.4 Issuance of certificates

A certificate obtained shall be issued in accordance with Article: "Issue of certificate" of Annex A and Article 6 of Annex B. Certificates issued by STIPEL shall comply with definition 3.5 and the requirements contained in paragraphs 9.4.7, 9.4.8 and 9.4.9 of ISO 17024:2012.

16.5 Suspension or revocation

A certificate that has been obtained may only be suspended or revoked by STIPEL if the requirements of Article 15 of Annex B are met.

[Source: ISO 17024:2012 par. 9.5]

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



17. Registration of certificates

- The certificates issued under the scope of this scheme are registered by STIPEL in a digital application created, maintained and managed by STIPEL for this purpose.
- Every certificate that has been issued contains a personalized QR-code that can be scanned.
- The certified person is automatically redirected to the digital application and the registered certificate by scanning this QR-code. Therein they can check the status and the period of validity of the certificate.

[Invulling van ISO 17024:2012 par. 7.1.1 en 7.2.1]

18 Procedures for insight, complaints, objections and appeals

Note 1 to this chapter: Appendix D contains a schematic overview of the categorization and handling of questions, requests, and comments.

18.1.1 Request for insight

- The examiner or surveyor can't give insight into the undertaken exam or do a postdiscussion of the undertaken exam to the candidate due to the strict confidentiality of the exam questions.
- A candidate only has a right for insight of a theoretical or practical exam by following the procedure for complaints of paragraph 14.2 or by filing a complaint with the examination body were the exam was undertaken.
- This complaint must specifically refer to the questions about which the candidate has remarks.

18.1.2 Assessment of the request for insight

- The examination body is qualified to asses if a request of insight is justified when the request is in regards to the insight of a practical exam. STIPEL is exclusively qualified to asses if a request is justified when the request is for the inspection of a theoretical exam.
- When this situation applies, STIPEL delegates the facilitation of the inspection to the concerned examination body.

The examination body and STIPEL can only grant a request of insight if:

- The request proves that the candidate has a complaint about the contents or assessment of one or more exam questions.

18.1.3 The facilitation of insight

- The examination body where the candidate has undertaken the exam is responsible for determining the day on which the insight will be given and facilitated, if a request for insight is granted in accordance with paragraph 14.1.2.
- The examination body selects a responsible person who shall ensure that:
 - the candidate is exclusively given insight into the exam-questions for which the request of insight of paragraph 14.1.2 was granted;

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



- the insight is undertaken in a room where only the candidate and the responsible person are located;
- the candidate doesn't make pictures of the exams, doesn't undertake fraudulent activities or tries to get insider knowledge into the exam-questions that he has done.

Remark one of this paragraph: The insight must be done in person.

18.2 Procedure for complaints

18.2.1 Grounds for filing a complain

Stakeholders may file a complaint in accordance with paragraph 18.2.2 if the complaint concerns:

- The content, development process, or modification of the central scheme and the associated profile documents, which serve as the basis for the assessment of candidates and certificate holders.

Applicants, candidates, and certificate holders may file a complaint in accordance with paragraph 18.2.3 if the complaint concerns:

- The services provided by STIPEL during the certification process.
- A request for access in accordance with paragraph 18.1.

Applicants, candidates, and certificate holders may file a complaint in accordance with paragraph 18.2.4 if the complaint concerns:

- The services provided by examination bodies during the certification process.

[Source ISO 17024:2012 par. 9.9]

18.2.2 Procedure for complaints about the STIPEL standards

Complaints regarding STIPEL standards may be submitted by completing the contact form available at <u>https://stipel.nl/contact/</u>. No submission deadline applies to this type of complaint.

To be considered for review, the complaint must include the following information:

- Surname and first name or initials of the complainant;
- The specific aspect of STIPEL standard to which the complaint relates: central scheme or associated profile documents;
- A substantiated explanation of the complaint.
- No later than six weeks after receipt, the complaint will be formally reviewed by the secretary, and the complainant will be informed of the progress and the final decision regarding the complaint.

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



- If the complaint is not resolved within this period or to the satisfaction of the complainant, the complainant may file an appeal with the Appeals Committee in accordance with paragraph 18.4.2.

[Source: ISO 17024:2012 par. 9.8]

18.2.3 Procedure for complaints about the service provided by STIPEL

Complaints regarding services provided by STIPEL may be submitted by completing the contact form available at <u>https://stipel.nl/contact/</u>. No submission deadline applies to this type of complaint.

To be considered for review, the complaint must include the following information:

- Surname and first name or initials of the complainant;
- A substantiated explanation of the complaint.
- No later than six weeks after receipt, the complaint will be formally reviewed by the secretary, and the complainant will be informed of the progress and the final decision regarding the complaint.
- If the complaint is not resolved within this period or to the satisfaction of the complainant, the complainant may file an appeal with the Appeals Committee in accordance with paragraph 18.4.3.

[Source: ISO 17024:2012 par. 9.8]

18.2.4 Procedure for complaints about the service provided by examination bodies

Complaints regarding the services provided by an Examination Body may be submitted by contacting the Examination Body directly or by completing the contact form. The contact details of the Examination Bodies can be found at https://stipel.nl/direct-regelen/examenbureau-vinden/. The contact form is available at <u>https://stipel.nl/contact/</u>. No submission deadline applies to this type of complaint.

To be considered for review, the complaint must include the following information:

- Surname and first name or initials of the complainant;
- A substantiated explanation of the complaint.
- No later than six weeks after receipt, the complaint will be formally reviewed by the secretary, and the complainant will be informed of the progress and the final decision regarding the complaint.
- If the complaint is not resolved within this period or to the satisfaction of the complainant, the complainant may file an appeal with the Appeals Committee in accordance with paragraph 18.4.4.

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



[Source: ISO 17024:2012 par. 9.8]

18.3 Procedure for objections

18.3.1 Grounds for Filing an Objection

Applicants, candidates, and certificate holders may file an objection in accordance with paragraph 18.3.2 if they have a complaint regarding a formal certification decision made by STIPEL concerning them, including but not limited to:

- The granting of a STIPEL certificate
- The suspension of a STIPEL certificate
- The withdrawal of a STIPEL certificate

Applicants, candidates, and certificate holders may file an objection in accordance with paragraph 18.3.3 if they have a complaint regarding a formal certification decision made by an Examination Body concerning them, including but not limited to:

- Admission to the certification process
- Assigned examination results

[Source: ISO 17024:2012 par. 9.8]

18.3.2 Procedure for objections STIPEL

An objection may be submitted by completing the objection form, which can be found at <u>https://stipel.nl/bezwaar/</u>. No submission deadline applies to this objection.

To be eligible for review, the form must include at least the following information:

- The name, initials, and place of birth of the objector;
- The certification application to which the objection relates;
- The reasons for the objection.
- No later than six weeks after receipt of the objection, the secretary must issue a reasoned decision on the objection and inform the objector of this decision in writing.
- If the objection is not resolved within this period or to the satisfaction of the objector, the objector may lodge an appeal with STIPEL in accordance with paragraph 18.4.5.

[Source ISO 17024:2012 par. 9.8]

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



18.3.3 Procedure for objections examination bodies

An objection can be submitted by contacting the Examination body or by completing the objection form. The contact details of the Examination bodies can be found by consulting https://stipel.nl/direct-regelen/examenbureau-vinden/. The objection form is available at https://stipel.nl/direct-regelen/examenbureau-vinden/. The objection form is available at https://stipel.nl/direct-regelen/examenbureau-vinden/. The objection form is available at https://stipel.nl/direct-regelen/examenbureau-vinden/. The objection form is available at https://stipel.nl/bezwaar/. No submission deadline applies to this objection.

In order for the objection to be processed, the objector must provide the following information to the Examination body or fill it in on the objection form:

- Last name and first name or initials of the objector;
- The requested certification to which the objection relates;
- The dates on which the exams were taken;
- The reasons for the objection.
- No later than six weeks after receipt of the objection, the Examination Board must provide a reasoned decision regarding the objection and inform the objector in writing.
- If the objection is not resolved within the deadline or to the satisfaction of the objector, the objector may appeal to STIPEL in accordance with paragraph 18.4.6.

[Source: ISO 17024:2012 par. 9.8]

18.4 Procedure for appeals

18.4.1 Grounds for submitting an appeal

Stakeholders may submit an appeal in accordance with paragraph 18.4.2 if the appeal concerns:

- A written decision made by STIPEL regarding a submitted complaint as referred to in paragraph 18.2.2.

Applicants, candidates and certificate holders may submit an appeal in accordance with paragraph 18.4.3 if the appeal concerns:

- A written decision made by STIPEL regarding a submitted complaint as referred to in paragraph 18.2.3.

Applicants, candidates and certificate holders may submit an appeal in accordance with paragraph 18.4.4 if the appeal concerns:

- A written decision made by STIPEL regarding a submitted complaint as referred to in paragraph 18.2.4.

Applicants, candidates and certificate holders may submit an appeal in accordance with paragraph 18.4.5 if the appeal concerns:

- A written decision made by STIPEL regarding a submitted objection as referred to in paragraph 18.3.2.

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



Applicants, candidates and certificate holders may submit an appeal in accordance with paragraph 18.4.5 if the appeal concerns:

- A written decision made by STIPEL regarding a submitted objection as referred to in paragraph 18.3.2.

Applicants, candidates and certificate holders may submit an appeal in accordance with paragraph 18.4.6 if the appeal concerns:

- A written decision made by an examination body regarding a submitted objection as referred to in paragraph 18.3.3.

[Source: ISO 17024:2012 par. 9.8 en 9.9]

18.4.2 Procedure for appeals against a decision on a complaint about STIPEL standards

An appeal can be submitted by sending an email to info@stipel.nl within 15 working days of receiving the written decision on the complaint. In order for the appeal to be processed, the email must include a justification for submitting the appeal.

- Receipt of the appeal will be confirmed in writing within 10 working days of its submission.
- The appeal will be reviewed by an Appeals Committee. The Appeals Committee is established on an ad hoc basis for each appeal and consists of members of the body within the meeting structure, as described in Chapter 12 of STIPEL 10001:2024B, who possess expertise in the standard that is central to the appeal.
- In its review, the Appeals Committee will assess both the substantive handling of the complaint and the diligence of the process, against the applicable requirements and regulations of the certification scheme.
- If the Appeals Committee concludes that the complaint was substantively handled correctly and that the process was sufficiently diligent and in accordance with the applicable requirements and regulations, the appellant will be deemed to be in the wrong.
- In other cases, the Appeals Committee will refer the complaint back to the secretary with additional conditions.
- Decisions resulting from the appeal will be communicated to the appellant in writing within six weeks of receipt of the appeal. This period may be extended once by six weeks. If the review period is extended, the appellant will be informed of this before the original deadline expires.

[Source: ISO 17024:2012 par. 9.8 en 9.9]

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



18.4.3 Procedure for appeals against a decision on a complaint about the service provided by STIPEL

An appeal can be submitted by sending an email to info@stipel.nl within 15 working days of receiving the written decision on the complaint. In order for the appeal to be processed, the email must include a justification for submitting the appeal.

- Receipt of the appeal will be confirmed in writing within 10 working days of its submission.
- The appeal will be reviewed by the Board of STIPEL.
- If the Board concludes that the complaint was handled with sufficient diligence and in accordance with the applicable requirements and regulations, the appellant will be deemed to be in the wrong.
- In other cases, the Board will refer the complaint back to the secretary, possibly with additional conditions.
- Decisions resulting from the appeal will be communicated to the appellant in writing within six weeks of receipt of the appeal. This period may be extended once by six weeks. If the review period is extended, the appellant will be informed of this before the original deadline expires.

[Source: ISO 17024:2012 par. 9.8 en 9.9]

18.4.4 Procedure for appeals against a decision on a complaint about the service provided by examination bodies.

An appeal can be submitted by sending an email to info@stipel.nl within 15 working days of receiving the written decision on the complaint. In order for the appeal to be processed, the email must include a justification for submitting the appeal.

- Receipt of the appeal will be confirmed in writing within 10 working days of its submission.
- The appeal will be reviewed by the secretary.
- If the secretary concludes that the complaint was handled with sufficient diligence and in accordance with the applicable requirements and regulations, the appellant will be deemed to be in the wrong.
- In other cases, the secretary will refer the complaint back to the Examination Body, possibly with additional conditions.
- Decisions resulting from the appeal will be communicated to the appellant in writing within six weeks of receipt of the appeal. This period may be extended once by six weeks. If the review period is extended, the appellant will be informed of this before the original deadline expires.

[Source: ISO 17024:2012 par. 9.8 en 9.9]

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



18.4.5 Procedure for appeals against a decision made by STIPEL on a objection

An appeal can be submitted by sending an email to info@stipel.nl within 15 working days of receiving the written decision on the complaint. In order for the appeal to be processed, the email must include a justification for submitting the appeal.

- Receipt of the appeal will be confirmed in writing within 10 working days of its submission.
- The appeal will be reviewed by the Board of STIPEL.
- If the Board concludes that the complaint was handled with sufficient diligence and in accordance with the applicable requirements and regulations, the appellant will be deemed to be in the wrong.
- In other cases, the Board will refer the complaint back to the secretary, possibly with additional conditions.
- Decisions resulting from the appeal will be communicated to the appellant in writing within six weeks of receipt of the appeal. This period may be extended once by six weeks. If the review period is extended, the appellant will be informed of this before the original deadline expires.

[Source: ISO 17024:2012 par. 9.8 en 9.9]

18.4.6 Procedure for appeals against a decision made by an examination body on a objection

An appeal can be submitted by sending an email to info@stipel.nl within 15 working days of receiving the written decision on the complaint. In order for the appeal to be processed, the email must include a justification for submitting the appeal.

- Receipt of the appeal will be confirmed in writing within 10 working days of its submission.
- The appeal will be reviewed by the secretary.
- If the secretary concludes that the complaint was substantively handled, with sufficient diligence and in accordance with the applicable requirements and regulations, the appellant will be deemed to be in the wrong.
- In other cases, the secretary will refer the complaint back to the Examination Body, possibly with additional conditions.
- Decisions resulting from the appeal will be communicated to the appellant in writing within six weeks of receipt of the appeal. This period may be extended once by six weeks. If the review period is extended, the appellant will be informed of this before the original deadline expires.

[Source: ISO 17024:2012 par. 9.8 en 9.9]

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025

_ _



Appendix A: Registration form

To be filled in by the examination body:

Name examination body:	
Applied for certification (requested certification based on STIPEL 10003):	
Initial certification/Recertification (cross out what doesn't apply)	

To be filled in by the candidate:

Name:	
Initials:	
Place of birth:	
Country of birth:	
Date of birth:	
Personal email-address:	

Request for accommodation of special needs
Take note: Evidence must be handed over to the examination body when modified examination is applied
Authorization registration in public register
Take note: no authorization limits the use of the certificate
Authorization registration in third-party systems (digital security passports) Take note: no authorization limits the use of the certificate
Exclusive issuance of the certificate to the examination body
Duplicate of the certificate to the trainer
Duplicate of the certificate to the employer

The optimal choices are displayed above. Alternate choices can be encircled. This has possible ramifications for the usefulness of the certificate.

Terms & Conditions and the certification scheme

Examination and certification are subject to the Terms & Conditions and the certification scheme. If a certificate is issued to the applicant by STIPEL, the terms of this registration form shall be deemed part of a certification agreement.

Signature applicant:

Date:

Signature:

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



Appendix B: Terms & Conditions

Definitions

Article 1

- 1. *Applicant*: person who has submitted an application to be admitted into the certification process and has signed the registration form in order to become a certified person
- 2. *Candidate*: applicant who has fulfilled specified prerequisites and has been admitted to the certification process.
- 3. *Certified person*: candidate who has passed the assessment and examination and has received a STIPEL certificate.

Applicability

Article 2

The Terms & Conditions apply to all agreements established between a person and STIPEL on the basis of an registration form or the issuance of a certificate to that person.

Certification requirements

Article 3

A certificate may be issued by STIPEL if the candidate:

- is 18 years old or older;
- has met all prerequisites of the certification scheme and relevant profile documents;
- has passed all parts of the examination that is required by the certification scheme and relevant profile documents.

Age

Article 4

- 1. Candidates may be admitted to the examination from the time they reach the age of 17. The candidate is entitled to the examination result. The certificate is issued with a validity date, that starts when the candidate reaches the age of 18.
- 2. The period of validity of a certificate of a person who has not yet reached the age of 18 has a different period of validity, due to the fact that the period of validity must be calculated from the time the certification decision is made.

Prerequisites

Article 5

The applicant must meet all prerequisites in accordance with the certification scheme and relevant profile documents.

Decisions on certification

Article 6

STIPEL must make a positive decision on certification within 10 business days after the candidate has passed the last part of examination and if the candidate has complied to the certification requirements in accordance with article 3.

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



Issuance of the certificate

Article 7

- 1. The certificate will be exclusively issued in a digital form by STIPEL. The following alternative forms are explicitly excluded: paper certificates and stickers for the purpose of a security passport.
- 2. If the candidate is eligible to receive a certificate after going through the certification process, the certificate will be issued by STIPEL to the personal e-mail address provided by the candidate, unless the candidate has agreed to an alternative e-mail address in the certification agreement.
- 3. The alternative e-mail address mentioned in the second paragraph can only be an e-mail address of an examination body that wishes to temporarily keep a certificate in its possession because they have concerns that the examination fees, not including any training fees, will not be paid.

Period of validity

Article 8

- 1. The period of validity of a certificate is three years, unless article 4(2) applies.
- **2.** A certified person must be admitted to recertification in accordance with the certification scheme and relevant profile documents

Use of the certificate

Article 9

The certified person may use the certificate to prove that he has the competences specified in the certificate, as long as the validity period of the certificate has not expired, the certificate has not been withdrawn and has not been suspended.

Obligations of the applicant

Article 10

- 1. The applicant must meet the admission requirements in accordance with the certification scheme and relevant profile documents and must prove that he has met them.
- 2. The applicant must cooperate in the preliminary investigation of the examination body and must provide the documents required for this purpose.

Obligations of the candidate

Article 11

- 1. The candidate must not in any way cheat or have prior knowledge of the examination assignments or of the examinations he is to take;
- 2. The candidate must cooperate with the examination and assessment of the examination body.
- 3. The candidate must know and follow the examination regulations of the examination body that have been approved by STIPEL.

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



Obligations of the certified person

Article 12

- 1. From the moment the certificate is received and during the validity period of the certificate, the certified person must comply with the requirements of the certification scheme and relevant profile documents.
- 2. The certified person may not suggest to third parties that STIPEL is responsible for the actions and activities of the certified person.
- 3. The certified person must notify STIPEL in writing as soon as he no longer complies with the certification conditions, the Terms & Conditions, the certification scheme and the relevant profile documents.

Confidentiality

Article 13

- 1. The applicant, candidate and certified person are required to keep information about the exam confidential and are forbidden to share information about the contents of the exams with third parties.
- 2. The provisions of paragraph 1 do not apply to information that:
 - is public knowledge or becomes public knowledge without unlawful action by the applicant, candidate and certified person;
 - can be proven to have been lawfully in the possession of the applicant, candidate and certified person;
 - has been published or disclosed to the relevant authority by virtue of a legal obligation, or a duty of care imposed upon the applicant, candidate and certified person.

Registration of personal data

Article 14

STIPEL keeps records of the following personal data of the applicant, candidate and certified person:

- last name and first name or initials;
- date of birth;
- birthplace;
- country of birth;
- personal email address;
- the alternative e-mail address, if any, to which the certificate is to be issued;
- personal STIPEL number;
- training details;
- exam work;
- exam results.

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



Privacy agreements

Article 15

- 1. STIPEL may collect, store or process personal data, obtained training data, examination work and examination results to go through the certification process and verify the obtained certification. STIPEL may disclose this data, which is subject to confidentiality, to qualified third parties for quality purposes. STIPEL may use the data as displayed on the certificate, with the exception of contact information and underlying exam results, to verify the validity of a certificate upon inquiry by third parties.
- 2. STIPEL may disclose the information as displayed on the certificate, with the exception of contact information and underlying examination results, in a register of certification that is owned by STIPEL.
- 3. The applicant may request the data mentioned in article 15 (2) not to be revealed and thereby accepts that the validity of their certificate is less easily verified by third parties and, in that case, indemnifies STIPEL from the possible negative consequences.
- 4. STIPEL may exchange the data as shown on the certificate, with the exception of contact information and underlying exam results, and have it processed by third parties to demonstrate the validity of the certificate through the systems and applications of these third parties as marketed for the purpose of effective communication between clients, employers and employees.
- 5. The applicant may request that the certificate data mentioned in article 15 (4) cannot be exchanged with third parties and thereby accepts that the validity of their certificate is less easily verified by clients, employers and employees. In this case, the applicant indemnifies STIPEL from the possible negative consequences.
- 6. The applicant may invoke the right to be forgotten and may request STIPEL to delete the data as mentioned in article 15 (1), with the exception of the certificate number, STIPEL number and the indication of the scope of certification. In this case, the certificate is withdrawn and its validity expires because the identity of the certified person can no longer be validated.

Withdrawal and suspension of the certificate

Article 16

- 1. STIPEL has the right to suspend an issued certificate if:
 - The certified person no longer meets one or more of their obligations of the Terms & Conditions.
 - The certified person no longer meets one or more of their obligations of the certification agreement.
 - The certified person does not meet their financial or other obligations of the examination agreement with the examination body.
 - The certified person no longer meets the certification criteria of the certification scheme and relevant profile documents.
- 2. STIPEL will notify the certified person in writing, when it decides to suspend an issued certificate, stating the grounds for the suspension. The obligation to inform the certified person is waived when the candidate has waived the provision of a personal e-mail address at the time of application.

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025



- 3. STIPEL has the right to withdraw an issued certificate if the grounds for suspension are not removed within three weeks.
- 4. A certificate is withdrawn in all cases if:
 - The certification agreement has been terminated.
 - The certified person dies.
- 5. When a certificate is suspended, the certified person loses their right to use the certificate to prove that they have the competencies specified in the certificate.
- 6. STIPEL must lift the suspension if the grounds for suspension are removed within three weeks.
- 7. When a certificate is withdrawn, the certified person must destroy the certificate and confirm such destruction in writing within 14 days of STIPEL's request.

Complaints, Objections, and Appeals Procedure

Article 17

Applicants, candidates, certificate holders, and stakeholders may submit complaints, file objections, and lodge appeals in accordance with STIPEL 10001:2025 or the most recent applicable version of this document.

Liability

Article 18

- 1. Under no circumstances shall STIPEL be liable for any form of consequential damages, including, but not limited to, damages suffered as a result of third party claims against the applicant, candidate and certified person who used a STIPEL certificate.
- 2. Under no circumstances is STIPEL liable for damages suffered by the applicant, candidate and certified person caused by third parties that are used by STIPEL in the execution of its obligations towards the applicant, candidate and certified person.
- 3. The limitations of liability of this Article shall not apply if the damage was caused by intent or gross negligence of STIPEL.

Modification clause

Article 19

- 1. STIPEL is entitled to modify the Terms & Conditions during the term of the certification agreement.
- 2. STIPEL is required to notify the certified person of any changes to the general terms and conditions.
- 3. If the certified person does not accept these changes, the certified person may terminate the certification agreement on the date the changes come into effect, which will result in the withdrawal of the certificate as described in Article 16 (4).

Duration and termination of certification agreements

Article 20

The certification agreement terminates automatically if the certificate is withdrawn.

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025

Annex C: Management of questions, requests, and comments

		Managemen	Management of questions, requests, and comments	nd comments		
Category		Complaints		Objections	ions	Informal
Subject	about STIPEL standards	about the service of an examination body	about the service of STIPEL	Against decision made by an examination body	Against decision made by STIPEL	Other
Initial handling by:	Stipel-secretary	Administration of the Examination Body	Stipel-secretary	Administration of the Examination Body	Stipel-secretary	Stipel-secretary
Appeals by:	STIPEL-ad hoc experts (CvS, EC, EB)	Stipel-secretary	STIPEL-board	STIPEL-secretary	STIPEL-board	None

Documentcode and version number	Implementation date
STIPEL 10001:2025	18-06-2025